# Effects of antenatal diet and physical activity on maternal and fetal outcomes: individual patient data meta-analysis and health economic evaluation

Ewelina Rogozińska, 1,2 Nadine Marlin, 3 Louise Jackson,<sup>4</sup> Girish Rayanagoudar,<sup>1</sup> Anneloes E Ruifrok,<sup>5,6</sup> Julie Dodds,<sup>1</sup> Emma Molyneaux,<sup>7</sup> Mireille NM van Poppel, 8,9 Lucilla Poston, 10 Christina A Vinter, 11 Fionnuala McAuliffe, 12 Jodie M Dodd, 13,14 Julie Owens, 13 Ruben Barakat, 15 Maria Perales, 15 Jose G Cecatti, 16 Fernanda Surita, 16 SeonAe Yeo, 17 Annick Bogaerts, 18,19 Roland Devlieger, 20 Helena Teede,<sup>21</sup> Cheryce Harrison,<sup>21</sup> Lene Haakstad,<sup>22</sup> Garry X Shen,<sup>23</sup> Alexis Shub,<sup>24</sup> Nermeen El Beltagy,<sup>25</sup> Narges Motahari,<sup>26</sup> Janette Khoury,<sup>27</sup> Serena Tonstad,<sup>27</sup> Riitta Luoto,<sup>28</sup> Tarja I Kinnunen,<sup>29</sup> Kym Guelfi,<sup>30</sup> Fabio Facchinetti,31 Elisabetta Petrella,31 Suzanne Phelan,<sup>32</sup> Tânia T Scudeller,<sup>33</sup> Kathrin Rauh, 34,35 Hans Hauner, 34 Kristina Renault, 11,36 Christianne JM de Groot,<sup>6</sup> Linda R Sagedal,<sup>37</sup> Ingvild Vistad,<sup>37</sup> Signe Nilssen Stafne,<sup>38,39</sup> Siv Mørkved,<sup>38,39</sup> Kjell Å Salvesen,<sup>40,41</sup> Dorte M Jensen, 42 Márcia Vitolo, 43 Arne Astrup, 44 Nina RW Geiker, 45 Sally Kerry, 3 Pelham Barton, 4 Tracy Roberts,<sup>4</sup> Richard D Riley,<sup>46</sup> Arri Coomarasamy,<sup>47</sup> Ben Willem Mol,<sup>48</sup> Khalid S Khan<sup>1,2</sup> and Shakila Thangaratinam, 1,2\* on behalf of the International Weight Management in Pregnancy (i-WIP) **Collaborative Group** 

- <sup>1</sup>Women's Health Research Unit, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, UK
- <sup>2</sup>Multidisciplinary Evidence Synthesis Hub, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, UK
- <sup>3</sup>Pragmatic Clinical Trials Unit, Blizard Institute, Barts and the London School of Medicine and Dentistry, London, UK
- <sup>4</sup>Health Economics Unit, School of Health and Population Sciences, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK
- <sup>5</sup>Department of Obstetrics and Gynecology, Academic Medical Centre, Amsterdam, the Netherlands
- <sup>6</sup>Department of Obstetrics and Gynaecology, Faculty of Medicine, VU University Medical Center, Amsterdam, the Netherlands
- <sup>7</sup>Section of Women's Mental Health, Health Service and Population Research Department, Institute of Psychiatry, King's College London, London, UK
- <sup>8</sup>Department of Public and Occupational Health, EMGO Institute for Health and Care Research (EMGO+), VU University Medical Center, Amsterdam, the Netherlands
- <sup>9</sup>Institute of Sport Science, University of Graz, Graz, Austria
- <sup>10</sup>Division of Women's Health, Women's Health Academic Centre, King's College London, St Thomas' Hospital, London, UK
- <sup>11</sup>Department of Obstetrics and Gynecology, Odense University Hospital, University of Southern Denmark, Odense, Denmark
- <sup>12</sup>School of Medicine & Medical Science, UCD Institute of Food and Health, Dublin, Ireland
- <sup>13</sup>The Robinson Research Institute, School of Medicine, Department of Obstetrics & Gynaecology, University of Adelaide, SA, Australia
- <sup>14</sup>Women's and Children's Health Network, Women's and Babies Division, North Adelaide, SA, Australia
- <sup>15</sup>Facultad de Ciencias de la Actividad Física y del Deporte, Universidad Politecnica de Madrid, Madrid, Spain
- <sup>16</sup>Department of Obstetrics and Gynecology, School of Medical Sciences, University of Campinas, Campinas, Brazil
- <sup>17</sup>School of Nursing, University of North Carolina at Chapel Hill, Chapel Hill, NC. USA
- <sup>18</sup>Research Unit Healthy Living, University Colleges Leuven-Limburg, Leuven, Belgium
- <sup>19</sup>Centre for Research and Innovation in Care, University of Antwerp, Antwerp, Belgium
- <sup>20</sup>Division of Mother and Child, Department of Obstetrics and Gynaecology, University Colleges Leuven-Limburg, Hasselt and University Hospitals KU Leuven, Leuven, Belgium
- <sup>21</sup>Monash Centre for Health Research and Implementation, School of Public Health, Monash University, Melbourne, VIC, Australia
- <sup>22</sup>Norwegian School of Sport Sciences, Department of Sports Medicine, Oslo, Norway
- <sup>23</sup>Department of Internal Medicine, University of Manitoba, Winnipeg, MB, Canada
- <sup>24</sup>Department of Obstetrics and Gynaecology, University of Melbourne, Melbourne, VIC, Australia

- <sup>25</sup>Department of Obstetrics and Gynecology, Alexandria University, Alexandria, Egypt
- <sup>26</sup>Department of Sport Physiology, Faculty of Physical Education and Sport Sciences, Mazandaran University, Babolsar, Iran
- <sup>27</sup>Department of Obstetrics and Gynecology, Oslo University Hospital, Oslo, Norway
- <sup>28</sup>UKK Institute for Health Promotion Research, Tampere, Finland
- <sup>29</sup>School of Health Sciences, University of Tampere, Tampere, Finland
- <sup>30</sup>School of Sport Science, Exercise and Health, University of Western Australia, Perth, WA, Australia
- <sup>31</sup>Mother-Infant Department, University of Modena and Reggio Emilia, Modena, Italy
- <sup>32</sup>Kinesiology Department, California Polytechnic State University, San Luis Obispo, CA, USA
- <sup>33</sup>Department of Management and Health Care, São Paulo Federal University, Santos, Brazil
- <sup>34</sup>Else Kröner-Fresenius-Center for Nutritional Medicine, Technische Universität München, Munich, Germany
- <sup>35</sup>Competence Centre for Nutrition, Freising, Germany
- <sup>36</sup>Departments of Obstetrics and Gynecology, Hvidovre Hospital, University of Copenhagen, Copenhagen, Denmark
- <sup>37</sup>Department of Obstetrics and Gynecology, Sorlandet Hospital Kristiansand, Kristiansand, Norway
- <sup>38</sup>Department of Public Health and General Practice, Faculty of Medicine, Norwegian University of Science and Technology, Trondheim, Norway
- <sup>39</sup>Clinical Services, St Olavs Hospital, Trondheim University Hospital, Trondheim, Norway
- <sup>40</sup>Department of Obstetrics and Gynaecology, Clinical Sciences, Lund University, Lund, Sweden
- <sup>41</sup>Department of Laboratory Medicine Children's and Women's Health, Faculty of Medicine, Norwegian University of Science and Technology, Trondheim, Norway
- <sup>42</sup>Department of Endocrinology, Odense University Hospital, Odense, Denmark
- <sup>43</sup>Department of Nutrition and the Graduate Program in Health Sciences, Federal University of Health Sciences of Porto Alegre, Porto Alegre, Brazil
- <sup>44</sup>Department of Nutrition, Exercise and Sports, University of Copenhagen, Copenhagen, Denmark
- <sup>45</sup>Nutritional Research Unit, Copenhagen University Hospital Herlev, Copenhagen, Denmark
- <sup>46</sup>Research Institute for Primary Care and Health Sciences, Keele University, Keele, UK
- <sup>47</sup>School of Clinical and Experimental Medicine, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK
- <sup>48</sup>The South Australian Health and Medical Research Institute, Adelaide, SA, Australia

<sup>\*</sup>Corresponding author s.thangaratinam@gmul.ac.uk

**Declared competing interests of authors:** Hans Hauner reports grants from the German Ministry of Education and Research, the Bavarian Ministry of Agriculture and Nutrition, the Bavarian Ministry of Health, the Helmholtz Center Munich, the Else Kröner-Fresenius Foundation, AOK Bavaria (health insurance fund), Amway and the German Research Foundation outside the submitted work. Ben Willem Mol reports other grants from ObsEva during the conduct of the study.

Published August 2017 DOI: 10.3310/hta21410

## **Plain English summary**

# Effects of antenatal diet and physical activity on maternal and fetal outcomes

Health Technology Assessment 2017; Vol. 21: No. 41

DOI: 10.3310/hta21410

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

Maternal obesity and excessive weight gain in pregnancy increase complications in the mother and baby. These may be reduced by diet and physical activity. It is possible that benefits are restricted to particular groups of women based on their body mass, age, number of previous children, ethnicity and underlying medical condition(s). We looked at the effects of diet and physical activity on weight gain in pregnancy and on the risk of complications in the mother and baby. We obtained anonymised data of individual participants from multiple studies, and combined them using the technique known as individual patient data meta-analysis. This was intended to allow us to identify particular groups of women who may benefit from diet and physical activity.

We established the International Weight Management in Pregnancy Collaborative Network, comprising anonymised data of 12,343 women from 36 studies. We found that diet, physical activity and mixed methods, individually and when analysed together, effectively reduced weight gain in pregnancy, possibly decreased complications in the mother and had no effect on the baby. The effects were similar in all groups of women.

We did not identify any benefit to mothers or their children when they gained weight within specific targets that are currently recommended in many countries. A mother's age and a history of previous births predicted weight gain in pregnancy.

Diet and physical activity in pregnancy reduced weight gain by 0.7 kg, and had no effect on combined complications. There were with no differences in these benefits between various groups of women. The rate of Caesarean section was reduced by the lifestyle intervention compared with usual care.

#### HTA/HTA TAR

### **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.236

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA** programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 12/01/50. The contractual start date was in February 2013. The draft report began editorial review in November 2015 and was accepted for publication in August 2016. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2017. This work was produced by Rogozińska et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### Health Technology Assessment Editor-in-Chief

**Professor Hywel Williams** Director, HTA Programme, UK and Foundation Professor and Co-Director of the Centre of Evidence-Based Dermatology, University of Nottingham, UK

#### **NIHR Journals Library Editor-in-Chief**

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

#### **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

**Professor Matthias Beck** Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

**Professor Geoffrey Meads** Professor of Health Sciences Research, Health and Wellbeing Research Group, University of Winchester, UK

**Professor John Norrie** Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

**Professor Martin Underwood** Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk